Author's response to reviews

Title: Adaptation of Chinese and English versions of the Ankylosing Spondylitis Quality of Life (ASQoL) scale for use in Singapore

Authors:

Ying Ying Leung (katyccc@hotmail.com; katy.leung.y.y@singhealth.com.sg)

Weixian Lee (lee.weixian@yahoo.com.sg)

Nai Lee Lui (Lui.nai.lee@singhealth.com.sg)

Matthew Rouse (mrouse@galen-research.com)

Stephen McKenna (SMcKenna@galen-research.com)

Julian Thumboo (julian.thumboo@singhealth.com.sg)

Version: 3 Date: 03 Aug 2017

Author’s response to reviews:

3 Aug 2017

Editor in-Chief
BMC Musculoskeletal Disorders

Sir,

Re: BMSD-D-17-00196 R2

Adaptation of Chinese and English versions of the Ankylosing Spondylitis Quality of Life (ASQoL) scale for use in Singapore.

Thank you for your letter dated 3 Aug 2017. Following the Reviewers’ comments and advice, we have revised our original manuscript. We are pleased to re-submit our revised manuscript and our point-by-point responses to the Reviewers’ advice and comments.

The revised manuscript has also been approved by all the authors and that they have given necessary attention to ensure the integrity of the work.

Thank you for your kind attention and we look forward to hearing your favorable reply.
Yours sincerely,

Dr. Ying-Ying Leung
Corresponding author
Department of Rheumatology & Immunology
Singapore General Hospital
Tel No.: (65) 63265276
Fax No.: (65) 62203321
E-Mail: katyccc@hotmail.com

Encl.

Point-to-point response to Reviewer(s)' advice and comments

1. Please add a short, brief Conclusion section after the Discussion section of your main manuscript, in order to adhere to journal submission requirements.
   • Reply: A conclusion section is added.

2. Please remove the phrase, "YYL (katyccc@hotmail.com) takes responsibility for the integrity of the work as a whole." All authors should be responsible for the work as a whole.
   • Reply: The statement is rephrased as “All authors take responsibility for the integrity of the work as a whole.”

3. Please upload all figures as separate figure files and remove them from the main manuscript. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.
   • Reply: Figures are uploaded as separate files. Figure captions are placed at the end of main manuscript in the “figure legends” session.

4. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.
   • Reply: TRACKED CHANGES OR HIGHLIGHTING ARE REMOVED.